% EJC papers *must* begin with the following two lines.
\documentclass[12pt]{article}
\usepackage{e-jc}
\specs{P3.8}{21(3)}{2014}

\usepackage{amsthm,amsmath,amssymb}
\usepackage{graphicx}
\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}
\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}

% all overfull boxes must be fixed;
% i.e. there must be no text protruding into the margins


% declare theorem-like environments
\theoremstyle{plain}
\newtheorem{theorem}{Theorem}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{fact}[theorem]{Fact}
\newtheorem{observation}[theorem]{Observation}
\newtheorem{claim}[theorem]{Claim}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{open}[theorem]{Open Problem}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{question}[theorem]{Question}

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{note}[theorem]{Note}

\def\Z{{\mathbb Z}}
\def\Stab{{\rm Stab}}
\def\fix{{\rm fix}}
\def\la{\langle}
\def\ra{\rangle}
\def\AGL{{\rm AGL}}
\def\tl{\triangleleft}
\def\Aut{{\rm Aut}}
\def\CI{{\rm CI}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

% if needed include a line break (\\) at an appropriate place in the title

\title{\bf On the Cayley isomorphism problem\\ for Cayley objects of nilpotent groups\\ of some orders}

% input author, affilliation, address and support information as follows;
% the address should include the country, and does not have to include
% the street address

\author{Edward Dobson\\
\small Department of Mathematics and Statistics\\[-0.8ex]
\small Mississippi State University\\[-0.8ex]
\small PO Drawer MA Mississippi State, MS 39762 U.S.A\\
\small\tt dobson@math.msstate.edu\\
\small and\\
\small IAM\\[-0.8ex]
\small University of Primorska\\[-0.8ex]
\small 6000 Koper, Slovenia
}

% \date{\dateline{submission date}{acceptance date}\\
% \small Mathematics Subject Classifications: comma separated list of
% MSC codes available from http://www.ams.org/mathscinet/freeTools.html}

\date{\dateline{Feb 11, 2013}{Jul 4, 2014}{Jul 21, 2014}\\
\small Mathematics Subject Classifications: 05E18, 05C25, 20F18}

\begin{document}
\begin{thebibliography}{10}

\bibitem{Adam1967}
A.~\'Ad\'am. \newblock {\em Research problem 2-10}. \newblock J. Combin. Theory 2:393, 1967.

\bibitem{DixonM1996}
John~D. Dixon and Brian Mortimer. \newblock {\em Permutation groups}.  \newblock Graduate Texts in
  Mathematics, vol. 163, Springer-Verlag, New York, 1996.

\bibitem{Dobson2003a}
Edward Dobson. \newblock {\em On isomorphisms of abelian {C}ayley objects of certain
  orders}. \newblock Discrete Math. 266:203--215, 2003.

\bibitem{ElspasT1970}
Bernard Elspas and James Turner. \newblock {\em Graphs with circulant adjacency
  matrices} \newblock  J. Combinatorial Theory 9:297--307, 1970.

\bibitem{Gorenstein1968}
Daniel Gorenstein. \newblock \emph{Finite groups}. \newblock Harper \& Row Publishers, New York,
  1968.

\bibitem{Hall1976}
Marshall Hall, Jr. \newblock \emph{The theory of groups}. \newblock Chelsea Publishing Co., New
  York, 1976, Reprinting of the 1968 edition.

\bibitem{KaluzninK1976}
L.~A. Kalu{\v{z}}nin and M.~H. Klin. \newblock {\em Some numerical invariants of
  permutation groups}.  \newblock Latvi\u\i sk. Mat. E\v zegodnik 18:81--99, 1976.

\bibitem{KovacsM2009}
I.~Kov{\'a}cs and M.~Muzychuk. \newblock {\em The group {$\mathbb Z^2_p\times \mathbb Z_q$} is
  a {CI}-group}. \newblock Comm. Algebra 37:3500-3515, 2009.


\bibitem{Meldrum1995}
J.~D.~P. Meldrum. \newblock {\em Wreath products of groups and semigroups}. \newblock Pitman
  Monographs and Surveys in Pure and Applied Mathematics, vol.~74, Longman,
  Harlow, 1995.

\bibitem{Muzychuk1997}
Mikhail Muzychuk. \newblock {\em On \'{A}d\'am's conjecture for circulant graphs}
 \newblock  Discrete Math. 176:285--298, 1997.

\bibitem{Muzychuk1999}
Mikhail Muzychuk. \newblock {\em On the isomorphism problem for cyclic combinatorial objects}.
  \newblock Discrete Math. 197/198:589--606.

\bibitem{MuzychukP2011}
M.~Muzychuk and F.~Pakovich. \newblock {\em Jordan-{H}\"older theorem for imprimitivity
  systems and maximal decompositions of rational functions}. \newblock Proc. Lond. Math.
  Soc. 102:1--24, 2011.

\bibitem{Palfy1987}
P.~P. P{\'a}lfy. \newblock {\em Isomorphism problem for relational structures with a
  cyclic automorphism}. \newblock European J. Combin. 8:35--43, 1987.


\bibitem{Wielandt1964}
Helmut Wielandt, \emph{Finite permutation groups}. \newblock Translated from the German
  by R. Bercov, Academic Press, New York, 1964.

\end{thebibliography}


\maketitle

% E-JC papers must include an abstract. The abstract should consist of a
% succinct statement of background followed by a listing of the
% principal new results that are to be found in the paper. The abstract
% should be informative, clear, and as complete as possible. Phrases
% like "we investigate..." or "we study..." should be kept to a minimum
% in favor of "we prove that..."  or "we show that...".  Do not
% include equation numbers, unexpanded citations (such as "[23]"), or
% any other references to things in the paper that are not defined in
% the abstract. The abstract will be distributed without the rest of the
% paper so it must be entirely self-contained.

\begin{abstract}
We give a necessary condition to reduce the Cayley isomorphism problem for Cayley objects of a nilpotent or abelian group $G$ whose order satisfies certain arithmetic properties to the Cayley isomorphism problem of Cayley objects of the Sylow subgroups of $G$ in the case of nilpotent groups, and in the case of abelian groups to certain natural subgroups.  As an application of this result, we show that $\Z_q\times\Z_p^2\times\Z_m$ is a CI-group with respect to digraphs, where $q$ and $p$ are primes with $p^2 < q$ and $m$ is a square-free integer satisfying certain arithmetic conditions (but there are no other restrictions on $q$ and $p$).

  % keywords are optional
  \bigskip\noindent \textbf{Keywords:} Cayley object; Cayley graph; isomorphism; CI-group
\end{abstract}

\section{Introduction}

In 1967 \'Ad\'am \cite{Adam1967} conjectured that any two circulant
graphs of order $n$ are isomorphic if and only if they are
isomorphic by a group automorphism of $\Z_n$.  While \'Ad\'am's
conjecture was quickly shown to be false \cite{ElspasT1970}, the conjecture
nonetheless generated much interest in the following question:
Are two Cayley graphs of a group $G$ isomorphic if and only if
they are isomorphic by a group automorphism of $G$?  If so, we say
that $G$ is a {\bf CI-group with respect to graphs}.  This problem
naturally generalizes to any class of combinatorial objects (see \cite{Muzychuk1999} for several equivalent formulations of the precise definition of a combinatorial object).
Namely, is it true that two Cayley objects of a group $G$ in some
class ${\cal K}$ of combinatorial objects are isomorphic if and
only if they are isomorphic by a group automorphism of $G$?  If
so, we say that {\bf $G$ is a CI-group with respect to ${\cal K}$}.  If
$G$ is a CI-group with respect to {\it every} class of
combinatorial objects, we say that $G$ is a {\bf CI-group}.  In 1987,
P\'alfy \cite{Palfy1987} proved the following remarkable result:

\begin{theorem}\label{Paltool}
A group $G$ is a CI-group if and only if $\gcd(n,\varphi(n)) = 1$
or $n = 4$, where $\varphi$ is Euler's phi function.
\end{theorem}

While P\'alfy's result is quite powerful, it does not tell us
anything in general about isomorphisms between Cayley objects of a
group $G$ if $G$ is not a CI-group, other than there exists isomorphic Cayley
objects of $G$ which are not isomorphic by a group automorphism of
$G$. For such groups, we are then left with the question of if two
Cayley objects of $G$ are isomorphic, then what are the possible
isomorphisms between them?  This is sometimes known as the Cayley
isomorphism problem.  Usually, one would like the solution to this
question to be a (hopefully) short list $L$ of possible
isomorphisms.  That is, two Cayley objects of $G$ are isomorphic if
and only if they are isomorphic by a function in the list $L$. In
1999, Muzychuk \cite{Muzychuk1999} showed that if $G$ is a cyclic group of
order $n$ and for any distinct primes $p$ and $q$ dividing $n$ we
have that $q$ does not divide $p - 1$, then any two Cayley objects
of $G$ are isomorphic by an automorphism that can be found in a
natural way from isomorphisms of Cayley objects of prime-power
orders that divide $n$.  Thus Muzychuk reduced the Cayley
isomorphism problem for Cayley objects of cyclic groups of some orders to the
Cayley isomorphism problem for Cayley objects of cyclic groups of
prime-power orders.  In 2003, the author \cite{Dobson2003a}, found a
sufficient condition to extend Muzychuk's result to all abelian
groups (with the same order conditions), and showed this sufficient
condition was satisfied by some abelian groups.  In this paper, we
extend the author's earlier result to nilpotent groups, as well as to abelian groups with more general order conditions (Theorem \ref{main}).  Finally, as an application we will extend the list of CI-groups with respect to digraphs by showing that $\Z_q\times\Z_p^2\times\Z_m$ is a CI-group with respect to digraphs, where $p$ and $q$ are distinct primes with $p^2 < q$ and $m$ satisfies certain arithmetic conditions (Theorem \ref{app}).

Throughout this paper, $G$ is a finite group.  For group theoretic
terms not defined in this paper, see \cite{DixonM1996}.  We begin with
some definitions.

\begin{definition}
Let $G$ be a transitive group acting on $\Omega$.  Let $X$ be the
set of all complete block systems of $G$.  Define a partial order
on $X$ by ${\cal B}\preceq{\cal C}$ if and only if every block of
${\cal C}$ is a union of blocks of ${\cal B}$.  We define ${\cal B}\vert_C$ to be
the complete block system of $\Stab_G(C) = \{g\in G:g(C) = C\}$, the {\bf set-wise stabilizer of $C\in{\cal C}$}, consisting of all those
blocks of ${\cal B}$ that are contained in $C$, $C\in{\cal C}$, and remark that ${\cal B}\vert_C$ is a complete block system of $\Stab_G(C)$ in its action on $C$.  By $\fix_G({\cal B})$ we mean the subgroup of $G$ which fixes each block of ${\cal B}$ set-wise.  That is, $\fix_G({\cal B}) = \{g\in G:g(B) = B{\rm\ for\ all\ }B\in{\cal B}\}$.  We denote by $\Stab_G(x)$ the stabilizer of $x\in X$.  That is, $\Stab_G(x) = \{g\in G:g(x) = x\}$.  Finally, $g\in G$ induces a natural permutation $g/{\cal B}$ in $S_{\cal B}$, and we set $G/{\cal B} = \{g/{\cal B}:g\in G\}$.
\end{definition}

\begin{definition}
Let $n = \Pi_{i=1}^rp_i^{a_i}$ be the prime factorization of $n$ and define $\Omega:{\mathbb N}\mapsto{\mathbb N} $ by $\Omega(n) = \Sigma_{i=1}^ra_i$.  Setting $m = \Omega(n)$, we say a transitive group $G$
of degree $n$ is {\bf $m$-step imprimitive} if there exists a
sequence of complete block system ${\cal B}_0\prec{\cal B}_1\prec
\ldots \prec {\cal B}_m$. Note that ${\cal B}_0$ consists of singletons, while ${\cal B}_m$ consists of the entire set on which $G$ acts.  A complete block system ${\cal B}$ will
be said to be {\bf normal} if ${\cal B}$ is formed by the orbits
of a normal subgroup.  We will say that $G$ is {\bf normally
$m$-step imprimitive} if each ${\cal B}_i$, $0\le i\le m$, is
formed by the orbits of a normal subgroup of $G$.
\end{definition}

\section{The main tool}

In this section, we will give a sufficient condition that will imply that the Cayley isomorphism problem for nilpotent groups of certain orders can be reduced to the Cayley isomorphism problem for groups of prime-power order (Theorem \ref{main} and Corollary \ref{nilmain}), and for abelian groups with more general arithmetic conditions (Theorem \ref{main}).  That these results have implications for the Cayley isomorphism problem is established in Theorem \ref{bigtool}.  The following result is straightforward, and so its proof is omitted.

\begin{lemma}\label{blocktool}
Let $G_1,G_2\le S_n$ be transitive such that both $G_1$ and $G_2$
admit ${\cal B}$ as a complete block system.  Then $\la
G_1,G_2\ra$ admits ${\cal B}$ as a complete block system.
\end{lemma}

%\begin{proof}
%Let $g\in\la G_1,G_2\ra$.  Then $g = g_1g_2\cdots g_r$ where if
%$i\equiv 1\ (\mod 2)$ then $g_i\in G_1$ and if $i\equiv 0\ (\mod
%2)$ then $g_i\in G_2$.  We will show by induction on $r$ that
%$g(B)\in{\cal B}$ for every $B\in{\cal B}$.

%If $r = 1$, then this is trivial as ${\cal B}$ is a complete block
%system of $G_1$.  Assume then that $g_1\cdots g_r(B)\in{\cal B}$.
%Let $g_{r+1}\in G_i$, where $i = 1$ if $r + 1$ is odd and $i = 2$
%if $r + 1$ is even.  If $B\in{\cal B}$, then $g_{r+1}(B) = B'$ for
%some $B\in{\cal B}$.  Then
%$$g(B) = g_1\cdots g_rg_{r+1}(B) = g_1\cdots g_r(B') = B''\in{\cal B}$$
%\noindent by the induction hypothesis.  Whence $g(B)\in{\cal B}$
%for every $B\in{\cal B}$ and the result follows by induction.
%\end{proof}

%\begin{note}
%Let $G$ be a transitive group with complete block systems ${\cal
%B}\preceq{\cal C}$.  For $C\in{\cal C}$, we denote the set of all
%blocks of ${\cal B}$ that contained in $C$ by $[{\cal B}]\vert_C$.
%\end{note}

The following result is trivial after observing that the hypothesis implies that $\fix_G({\cal B}) = \Stab_G(B)$.

\begin{lemma}\label{mstep}
Let $G\le S_n$ be $m$-step imprimitive with
sequence ${\cal B}_0,\ldots,{\cal B}_m$.  If $G/{\cal B}_{m-1}$ is
cyclic of prime order $p$, then $\fix_G({\cal
B}_{m-1})\vert_B$ is $(m-1)$-step imprimitive for every
$B\in{\cal B}_{m-1}$, with $(m-1)$-step imprimitivity
sequence ${\cal B}_0\vert_B,\ldots,{\cal
B}_{m-1}\vert_B$.
\end{lemma}

%\begin{proof}
%We must show that $[{\cal B}_i]\vert_{B_{m-1}}$ is a complete
%block system of $\fix_G({\cal B}_{m-1})\vert_{B_{m-1}}$ for every
%$0\le i\le m-1$. It thus suffices to show that a block of ${\cal
%B}_i$, $0\le i\le m-1$, contained in $B_{m-1}$ is a block of
%$\fix_G({\cal B}_{m-1})\vert_{B_{m-1}}$. Let $a\in B_{m-1}$, and
%$B_i\in{\cal B}_i$ such that $a\in B_i$.  By \cite[Theorem
%1.1.5A]{DixonM1996}, there is a subgroup $H\ge\Stab_G(a)$ such that $B_i =
%\{h(a):h\in H\}$. Also by \cite[Theorem 1.1.5A]{DixonM1996}, we must show
%that there exists a subgroup $H_i\ge \Stab_{\fix_G({\cal
%B}_{m-1})\vert_{B_{m-1}}}(a)$ such that $B_i = \{h(a):h\in H_i\}$.
%Now, as $H$ fixes $B_i\subset B_{m-1}$ set-wise and $B_{m-1}$ is a
%block of $G$, $H$ fixes $B_{m-1}$ set-wise. As $G/{\cal B}_{m-1}$
%is cyclic of prime order $p$, we must have that $H\le\fix_G({\cal
%B}_{m-1})$. Similarly, as $a\in B_i\subset B_{m-1}$,
%$\Stab_G(a)\le\fix_G({\cal B}_{m-1})$. Hence $\Stab_G(a) =
%\Stab_{\fix_G({\cal B}_{m-1})}(a)$. Clearly then
%$\Stab_{\fix_G({\cal B}_{m-1})}(a)\vert_{B_{m-1}} =
%\Stab_G(a)\vert_{B_{m-1}} \le \Stab_{\fix_G({\cal
%B}_1)\vert_{B_1}}(a)$.  Conversely, if $g\in\Stab_{\fix_G({\cal
%B}_{m-1})}(a)\vert_{B_{m-1}}$, then there exists
%$g'\in\fix_G({\cal B}_1)$ such that $g'\vert_{B_{m-1}} = g$.  As
%$\Stab_G(a)\le\fix_G({\cal B}_{m-1})$, $g'\in\Stab_G(a)$. But then
%$g'\in\fix_G({\cal B}_{m-1})$ so that $g\in\Stab_{\fix_G({\cal
%B}_{m-1})\vert_{B_{m-1}}}(a)$. Hence $\Stab_{\fix_G({\cal
%B}_{m-1})}(a)\vert_{B_{m-1}} = \Stab_{\fix_G({\cal
%B}_{m-1})\vert_{B_{m-1}}}(a)$. Let $H_i = H\vert_{B_i}$.  As
%$\Stab_G(a)\le H\le\fix_G({\cal B}_{m-1})$ and $\Stab_G(a) =
%\Stab_{\fix_G({\cal B}_{m-1})}(a)$, we have that
%$\Stab_{\fix_G({\cal B}_{m-1})\vert_{B_{m-1}}}(a)\le
%H\vert_{B_{m-1}} = H_i$, and clearly $\{h_i(a):h_i\in
%H\vert_{B_{m-1}}\} = \{h(a):h\in H\}$ as ${\cal B}_i\prec {\cal
%B}_{m-1}$. The result is then proved.
%\end{proof}

We will use the following basic (and known) result implicitly throughout the
paper.

\begin{lemma}\label{regularabelian}
Let $G\le S_n$ be transitive with $H\le G$ a transitive abelian
subgroup.  Then every complete block system of $G$ is normal and
is formed by the orbits of a normal subgroup of $H$.
\end{lemma}

\begin{proof}
Let ${\cal B}$ be a complete block system of $G$ consisting of $m$
blocks of size $k$. As a transitive abelian group is regular
\cite[Proposition 1.4.4]{Wielandt1964}, we have that $H/{\cal B}$ is regular
of degree $m$, so that $\fix_H({\cal B})\not = 1$ and has order
$k$.  As $\Stab_H(B) = \fix_H({\cal B})$ for every $B\in{\cal B}$
and $\Stab_H(B)\vert_B$ is transitive \cite[Exercise 1.5.6]{DixonM1996},
we have that $\fix_H({\cal B})\vert_B$ is transitive for every
$B\in{\cal B}$.  As the blocks of ${\cal B}$ have size $k$, we
conclude that the orbits of $\fix_H({\cal B})\le\fix_G({\cal B})$
form ${\cal B}$.
\end{proof}

\begin{definition}
Let $G$ be a permutation group acting on $X$ and $H$ a permutation group acting on $Y$.  Define the {\bf wreath product of $G$ and $H$}, denoted $G\wr H$, to be the group of all permutations of $G\times H$ of the form $(x,y)\to (g(x),h_x(y))$.
\end{definition}

\begin{lemma}\label{mstepsoltool}
Let $n$ be a positive integer and $G_1,G_2$ be transitive abelian
groups of degree $n$ such that $\la G_1,G_2\ra$ is $m$-step
imprimitive.  Let $n = p_1^{a_1}p_2^{a_2}\cdots p_r^{a_r}$ be the
prime-power decomposition of $n$.  Then there exists $\delta\in\la
G_1,G_2\ra$ and a sequence of primes $q_1,\ldots,q_m$ such that $n
= q_1\cdots q_m$ and $\la G_1,\delta^{-1}G_2\delta\ra$ is
permutation isomorphic to a subgroup of
$\AGL(1,q_1)\wr(\AGL(1,q_2)\wr(\cdots\wr\AGL(1,q_m)))$.
Furthermore, if $\la G_1,G_2\ra$ is solvable, then we may take
$\delta = 1$.
\end{lemma}

\begin{proof}
We proceed by induction on $m$.  If $m = 1$, then $n$ is prime,
and both $G_1$ and $G_2$ are Sylow $n$-subgroups of $S_n$.  Hence
there exists $\delta\in\la G_1,G_2\ra$ such that
$\delta^{-1}G_2\delta = G_1$, and the result is trivial as $\la
G_1,\delta^{-1}G_2\delta\ra$ is cyclic of order $n$. Now assume that the result is true for all $m-1\ge 1$,
and let $G_1,G_2$ be transitive abelian groups of degree $n$,
where $\Omega(n) = m$, such that $\la G_1,G_2\ra$ is $m$-step
imprimitive.

As $\la G_1,G_2\ra$ is $m$-step imprimitive, $\la G_1,G_2\ra$ admits
a normal complete block system ${\cal B}$ consisting of $n/q_m$
blocks of size $q_m$ for some prime $q_m\vert n$, and both
$G_1/{\cal B}$ and $G_2/{\cal B}$ are transitive abelian groups of
degree $n/q_m$ and $\Omega(n/q_m) = m-1$. Furthermore, as $\la G_1,G_2\ra$
is $m$-step imprimitive, $\la G_1,G_2\ra/{\cal B}$ is $(m-1)$-step
imprimitive by \cite[Lemma 8]{Dobson2003a}, so by the induction hypothesis,
there exists $\delta_1\in\la G_1,G_2\ra$ such that $\la
G_1,\delta_1^{-1}G_2\delta\ra/{\cal B}$ is permutation isomorphic to
a subgroup of
$\AGL(1,q_1)\wr(\AGL(1,q_2)\wr(\cdots\wr\AGL(1,q_{m-1})))$ for some
sequence of primes $q_1,\ldots,q_{m-1}$ such that $n/q_m = q_1\cdots
q_{m-1}$, and if $\la G_1,G_2\ra$ is solvable, we may take $\delta_1
= 1$. Furthermore, $\fix_{G_1}({\cal B})$ is semiregular of order
$q_m$, and $\fix_{\delta_1^{-1}G_2\delta_1}({\cal B})$ is also
semiregular of order $q_m$. Hence there exists $\delta_2\in\fix_{\la
G_1,\delta_1^{-1}G_2\delta_1\ra}({\cal B})$ such that
$\delta_2^{-1}\fix_{\delta_1^{-1}G_2\delta_1}({\cal B})\delta_2$ is
contained in the same Sylow $q_m$-subgroup of $\fix_{\la
G_1,\delta_1^{-1}G_2\delta_1\ra}({\cal B})$ as $\fix_{G_1}({\cal
B})$.  If $\la G_1,G_2\ra$ is solvable, then $\fix_{\la
G_1,G_2\ra}({\cal B})$ is solvable, so $\fix_{\la G_1,G_2\ra}({\cal
B})\vert_B$ is solvable, and by \cite[Exercise 3.5.1]{DixonM1996},
$\fix_{\la G_1,G_2\ra}({\cal B})\vert_B$ has a unique Sylow
$q_m$-subgroup, so we may take $\delta_2 = 1$. Let $\delta =
\delta_1\delta_2$. As a Sylow $q_m$-subgroup of $\fix_{\la
G,\delta^{-1}G\delta\ra}({\cal B})$ is contained in
$1_{S_{n/q_m}}\wr\Z_{q_m}$ we have that both $G_1$ and
$\delta^{-1}G_2\delta$ normalize $1_{S_{n/q_m}}\wr\Z_{q_m}$. This
then implies that $\Stab_{\la
G_1,\delta^{-1}G_2\delta\ra}(B)\vert_B$ has a normal Sylow
$q_m$-subgroup, so that $\Stab_{\la
G_1,\delta^{-1}G_2\delta\ra}(B)\vert_B$ is permutation isomorphic to
a subgroup of $\AGL(1,q_m)$ for every $B\in{\cal B}$. It then
follows by the Embedding Theorem \cite[Theorem 2.6]{Meldrum1995}, that $\la
G_1,\delta^{-1}G_2\delta\ra$ is permutation isomorphic to a subgroup
of $\AGL(1,q_1)\wr(\AGL(1,q_2)\wr(\cdots\wr\AGL(1,q_m)))$, and the
result follows by induction.
\end{proof}

\begin{definition}
Let $\pi$ be a set of primes.  A {\bf $\pi$-group} $G$ is a group
such that every prime divisor of $\vert G\vert$ is contained in
$\pi$.  A subgroup $H$ of $G$ is an {\bf $S_\pi$-subgroup of $G$}
if no prime in $\pi$ divides $\vert G\vert/\vert H\vert$.  By
$\pi'$, we denote the set of primes dividing $\vert G\vert$ that
are not contained in $\pi$.
\end{definition}

We shall have need a consequence of the preceding  result.

\begin{lemma}\label{piabelian}
Let $n$ be a positive integer and $\pi$ be the set of distinct
prime numbers dividing $n$.  If $G_1,G_2$ are transitive abelian
groups of degree $n$ such that $\la G_1,G_2\ra$ is $m$-step
imprimitive then there exists $\delta\in\la G_1,G_2\ra$ such that
$\la G_1,\delta^{-1}G_2\delta\ra$ is a solvable $\pi$-group.
\end{lemma}

\begin{proof}
Let $n = p_1^{a_1}\cdots p_r^{a_r}$ be the prime-power
decomposition of $n$.  By Lemma \ref{mstepsoltool}, there exists
$\delta_1\in\la G_1,G_2\ra$ and a sequence $q_1,\ldots,q_m$ of
primes such that $n = q_1\cdots q_m$ and $\la
G_1,\delta_1^{-1}G_2\delta_1\ra\le\AGL(1,q_1)\wr(\AGL(1,q_2)\wr(\cdots\wr\AGL(1,q_m)))$.
As $\AGL(1,q_i)$ is solvable for every $1\le i\le m$, $\la
G_1,\delta_1^{-1}G_2\delta_1\ra$ is solvable.  By Hall's Theorem
\cite[Theorem 6.4.1]{Gorenstein1968}, we have that $G_1$ is contained in an
$S_\pi$-subgroup $H_1$ of $\la G_1,\delta_1^{-1}G_2\delta_1\ra$
and that $G_2$ is contained in an $S_\pi$-subgroup $H_2$ of $\la
G_1,\delta_1^{-1}G_2\delta_1\ra$.  Also by Hall's Theorem, there
exists $\delta_2\in\la G_1,\delta_1^{-1}G_2\delta_1\ra$ such that
$\delta_2^{-1}H_2\delta = H_1$.  Let $\delta = \delta_1\delta_2$.
Then $\la G_1,\delta^{-1}G_2\delta\ra\le H_1$, and $H_1$ is a
solvable $\pi$-group.
\end{proof}

Let ${\cal L} = {\cal L}_G$ be the set of all normal complete
block systems of a transitive group $G$.  Then $\preceq$ is a canonical partial order on ${\cal L}$.  Define operations $\cup$ and $\cap$ on
${\cal L}$ by ${\cal B}\cup{\cal C}$ is the normal complete block
system of $G$ formed by the orbits of $\la\fix_G({\cal
B}),\fix_G({\cal C})\ra = \fix_G({\cal B})\cdot\fix_G({\cal C})$
(as both of these groups are normal), and ${\cal B}\cap{\cal C}$
is the normal complete block system of $G$ formed by the orbits
of $\fix_G({\cal B})\cap\fix_G({\cal C})$. Notice that both of
these operation do in fact give normal complete block systems as
$\la\fix_G({\cal B}),\fix_G({\cal C})\ra\tl G$ and $\fix_G({\cal
G})\cap\fix_G({\cal C})\tl G$. Thus ${\cal L}_G$ is a lattice. See \cite{Hall1976} for terms regarding lattices not defined here.  We
also have that

\begin{lemma}\label{lattice}
If $G$ contains a transitive abelian group $H$, then ${\cal L}_G$
is a modular lattice.
\end{lemma}

\begin{proof}
We must show that if ${\cal B}\preceq{\cal A}$, then ${\cal
A}\cap({\cal B}\cup{\cal C}) = {\cal B}\cup({\cal A}\cap{\cal C})$.
By Lemma \ref{regularabelian}, there exists $A,B,C\le H$ such that ${\cal A}$ is formed by the
orbits of $A$, ${\cal B}$ is formed by the orbits of $B$, and ${\cal
C}$ is formed by the orbits of $C$.  As ${\cal B}\preceq{\cal A}$,
we have that $B\le A$.  By \cite[Theorem 8.4.1]{Hall1976}, we have that
$A\cap(B\cdot C) = B\cdot(A\cap C)$. Then the orbits of
$A\cap(B\cdot C)$ are the same as the orbits of $B\cdot(A\cap C)$.
\end{proof}

We remark that the previous result is contained in \cite[Theorem 2.10]{MuzychukP2011}

In the following three results, we will have in the hypothesis
that $\gcd(n_i,n_j\cdot\varphi(n_j)) = 1$.  Notice that this
implies that $\gcd(n_i,n_j) = 1$, and that if $p_i\vert n_i$ is
prime, then $p_i$ does not divide $p_j - 1$ for any prime
$p_j\vert n_j$.

\begin{lemma}\label{solproduct}
Let $n_1,\ldots,n_r$ be positive integers such that if $i\not =
j$, then $\gcd(n_i,n_j\cdot\varphi(n_j)) = 1$, $\pi_i$ the set of
primes dividing $n_i$, and $H_i$ be a transitive, solvable,
$\pi_i$-group of degree $n_i$, $1\le i\le r$. Let $G$ be a
transitive $m$-step imprimitive subgroup of $\Pi_{i=1}^rH_i$
acting coordinate-wise, where $\Omega(n_1\cdots n_r) = m$.
Then there exists transitive subgroups $L_i\le H_i$ such that $G =
\Pi_{i=1}^rL_i$.
\end{lemma}

%\begin{proof}
%It is not difficult to see that $\Pi_{i=1}^rH_i$ admits complete
%block systems ${\cal C}_i$ consisting of $n/n_i$ blocks of size
%$n_i$, $1\le i\le r$.  As each $H_i$ is solvable, we have that $G$
%is solvable, and so contains an $S_{\pi_i}$-subgroup $L_i$ for
%every $1\le i\le r$.  Notice that if we show $\fix_G({\cal C}_i) =
%L_i$, then we will have that $L_i\cap L_j = 1$ for every $i\not =
%j$, $L_i\tl G$, and $\la L_i:1\le i\le r\ra = G$. The result will
%then follow. Now, $G/{\cal C}_i$ is certainly solvable, and we
%have by \cite[Lemma 8]{Dobson2003a} that $G/{\cal C}_i$ is $m_i$-step
%imprimitive, where $n/n_i$ has exponent $m_i$. By Lemma
%\ref{mstepsoltool}, we have that there exists a sequence of primes
%$q_1,\ldots,q_{m_i}$ such that $q_1\cdots q_{m_i} = n/n_i$ and
%$G/{\cal C}_i\le
%\AGL(1,q_1)\wr(\AGL(1,q_2)\wr(\cdots\wr\AGL(1,q_{m_i})))$. As
%$\vert\AGL(1,p)\vert = (p-1)\cdot p$ and
%$\gcd(n_i,n_j\cdot\varphi(n_j)) = 1$ if $i\not = j$, we have that
%$G/{\cal C}_i$ has order relatively prime to $n_i$.  We conclude
%that $\fix_G({\cal C}_i) = L_i$, and the result follows.
%\end{proof}

\begin{proof}
It is not difficult to see that $\Pi_{i=1}^rH_i$ admits complete
block systems ${\cal C}_i$ consisting of $n/n_i$ blocks of size
$n_i$, $1\le i\le r$, formed by the orbits of $H_i$.  As each $H_i$ is solvable, we have that $G$
is solvable, and so contains an $S_{\pi_i}$-subgroup $L_i$ for
every $1\le i\le r$.  As $\Pi_{i=1}^rH_i/{\cal C}_i$ is a $\pi_i'$-group, $G/{\cal C}_i$ is also a $\pi_i'$-group, and so $\fix_G({\cal C}_i) = L_i$.  Then $L_i\cap L_j = 1$ for every $i\not =
j$, $L_i\tl G$, and $\la L_i:1\le i\le r\ra = G$. The result now follows.
\end{proof}

We now need only one more tool to prove the main result (Theorem \ref{main}) of this section. Before proceeding to this last tool, it will be useful to develop some terminology which will simplify the statement.  First, the proof of Theorem \ref{main} will proceed by induction on $m = \Omega(n)$.  So when proving Theorem \ref{main}, we will be assuming that the conclusion of Theorem \ref{main} holds for all integers $n/p$, where $p$ divides $n$ is prime.  In particular, with $m,n_1,\ldots,n_r$ and $n$ satisfying the hypothesis of Theorem \ref{main} and $G_1,G_2$ transitive abelian or nilpotent groups of degree $n$, then whenever $K_1,K_2$ are transitive nilpotent or abelian groups of degree $n/p$ - and to simplify our notation, there is no harm in assuming that $p\vert n_1$ - such that $\la K_1,K_2\ra$ are $(m-1)$-step imprimitive, then there exists $\delta\in\la K_1,K_2\ra$ such that $\la K_1,\delta^{-1}K_2\delta\ra\le \Pi_{i=1}^r\bar{H}_i$, where $\bar{H}_i$ is a solvable $\bar{\pi}_i$ group of degree $\bar{n}_i$.  Here $\bar{n}_i = n_i$ if $i\not = 1$ while $\bar{n}_1 = n_1/p$, and $\bar{\pi}_i$ is the set of prime divisors of $\bar{n}_i$.  In this situation, we will say that $n$ {\bf satisfies the main induction hypothesis}.

\begin{lemma}\label{bottomtool}
Let $n_1,\ldots,n_r$ be positive integers such that if $i\not =
j$, then $\gcd(n_i,n_j\cdot\varphi(n_j)) = 1$.  Let $n = n_1\cdots
n_r$ with $\Omega(n) = m$, $p\vert n$ (and without loss of generality, $p\vert n_1$), and $\bar{\pi} = \cup_{j\in
I}\bar{\pi}_j$ for some $I\subseteq[r]$. If
\begin{enumerate}
\item $n$ satisfies the main induction hypothesis,
\item $\la G_1,G_2\ra$ is $m$-step imprimitive,
\item $\la G_1,G_2\ra$ admits a complete block system ${\cal B}$ of $p$
blocks of size $n/p$, and
\item $\la G_1,G_2\ra/{\cal B}$ is a $p$-group,
\end{enumerate}
\noindent then there exists $\delta\in\la G_1,G_2\ra$ such that
$\la G_1,\delta^{-1}G_2\delta\ra$ admits a complete block system
formed by the orbits of the unique $S_{\bar{\pi}}$-subgroup of
$\fix_{G_i}({\cal B})$, $i = 1,2$.
\end{lemma}

\begin{proof}
Let $B\in{\cal B}$ and $K_i = \fix_{G_i}({\cal B})\vert_B$, $i = 1,2$. As $\la G_1,G_2\ra/{\cal B}$ has
prime order $p$, we must have that $\Stab_{\la G_1,G_2\ra}(B) =
\fix_{\la G_1,G_2\ra}({\cal B})$.  Note that $K_i$ is transitive on $B$, and if $G_i$ is nilpotent or abelian, then $K_i$ is nilpotent or abelian, $i = 1,2$.  Clearly $\fix_{G_1}({\cal
B}),\fix_{G_2}({\cal B})\le\fix_{\la G_1,G_2\ra}({\cal B})$.  By
Lemma \ref{mstep}, $\fix_{\la G_1,G_2\ra}({\cal B})\vert_B$ is
$(m-1)$-step imprimitive, so that $\la K_1,K_2\ra$ is $(m-1)$-step
imprimitive.

By hypothesis, there exists $\delta_1\in\la\fix_{G_1}({\cal
B}),\fix_{G_2}({\cal B})\ra$ such that

$$\la\fix_{G_1}({\cal
B}),\delta_1^{-1}\fix_{G_2}({\cal B})\delta_1\ra\vert_B\le
\Pi_{j=1}^r\bar{H}_{B,j},$$

\noindent where each $\bar{H}_{B,j}$ is a transitive solvable
$\bar{\pi}_j$-group of degree $\bar{n}_j$. Similarly, if $B\not = B'\in{\cal B}$, then there exists $\delta_2\in\la\fix_{G_1}({\cal
B}),\delta_1^{-1}\fix_{G_2}({\cal B})\delta_1\ra$ such that

$$\la\fix_{G_1}({\cal
B}),\delta_2^{-1}\delta_1^{-1}\fix_{G_2}({\cal
B})\delta_1\delta_2\ra\vert_{B'}\le \Pi_{j=1}^r\bar{H}_{B',j},$$ where
each $\bar{H}_{B',j}$ is a transitive solvable $\bar{\pi}_j$-group of degree
$\bar{n}_j$. Furthermore, we have that
$\delta_2\vert_B\in\la\fix_{G_1}({\cal
B}),\delta_1^{-1}\fix_{G_2}({\cal B})\delta_1\ra\vert_B$.  This
then implies that
$$\la\fix_{G_1}({\cal B}),\delta_2^{-1}\delta_1^{-1}\fix_{G_2}({\cal B})\delta_1\delta_2\ra\vert_B \le \la\fix_{G_1}({\cal B}),\delta_1^{-1}\fix_{G_2}({\cal B})\delta_1\ra\vert_B.$$
\noindent Hence $\la\fix_{G_1}({\cal
B}),\delta_2^{-1}\delta_1^{-1}\fix_{G_2}({\cal
B})\delta_1\delta_2\ra\vert_B\le \Pi_{j=1}^r\bar{H}_{B,j}$.  Continuing
inductively, we have that there exists $\delta\in\la\fix_{G_1}({\cal
B}),\fix_{G_2}({\cal B})\ra$ such that $\la\fix_{G_1}({\cal
B}),\delta^{-1}\fix_{G_2}({\cal
B})\delta\ra\vert_B\le\Pi_{j=1}^r\bar{H}_{B,j}$ for every $B\in{\cal B}$,
where each $\bar{H}_{B,j}$ is a transitive solvable $\bar{\pi}_j$-group of
degree $\bar{n}_j$. Note that $\delta^{-1}\fix_{G_2}({\cal B})\delta =
\fix_{\delta^{-1}G_2\delta}({\cal B})$ as $\delta/{\cal B} = 1$. For
ease of notation, we will replace $\delta^{-1}G_2\delta$ by $G_2$
and thus assume without loss of generality that $\la\fix_{G_1}({\cal
B}),\fix_{G_2}({\cal B})\ra\vert_B\le \Pi_{j=1}^rH_{B,j}$ for each
$B\in{\cal B}$.  By Lemma \ref{solproduct}, we may assume without
loss of generality that $\la\fix_{G_1}({\cal B}),\fix_{G_2}({\cal
B})\ra\vert_B = \Pi_{j=1}^r\bar{H}_{B,j}$ for each $B\in{\cal B}$.

As $\la\fix_{G_1}({\cal B}),\fix_{G_2}({\cal B})\ra\vert_B =
\Pi_{j=1}^rH_{B,j}$ for each $B\in{\cal B}$, $\la\fix_{G_1}({\cal
B}),\fix_{G_2}({\cal B})\ra\vert_B$ has a normal subgroup $L_B$
with orbits of size $\Pi_{j\in I}\bar{n}_j$, and so
$\la\fix_{G_1}({\cal B}),\fix_{G_2}({\cal B})\ra\vert_B$ admits a
complete block system ${\cal C}_B$ formed by the orbits of $L_B$,
$B\in{\cal B}$.  Also note that the $S_{\bar{\pi}}$-subgroup of
$\fix_{G_i}({\cal B})\vert_B$ must be contained in $L_B$, as
otherwise $(\fix_{G_i}({\cal B})\vert_B)/{\cal C}_B$ contains a
nontrivial normal subgroup with orbits of size dividing $\Pi_{j\in
I}\bar{n}_j$, $i = 1,2$. However, $(\fix_{G_i}({\cal
B})\vert_B)/{\cal C}_B$ is a solvable $\bar{\pi}'$-subgroup and
$\gcd(\Pi_{j\in I}\bar{n}_j,\Pi_{j\not\in I,j\in[r]}\bar{n}_j) = 1$, $i =
1,2$, a contradiction. Then $\fix_{G_i}({\cal B})\vert_B$, $i = 1,2$, admit
complete block systems ${\cal D}_B$ formed by the orbits of their
unique $S_{\bar{\pi}}$-subgroups, respectively, and these complete block
systems must be precisely ${\cal C}_B$, for $B\in{\cal B}$.

Let ${\cal C} = \cup_{B\in{\cal B}}{\cal C}_B$.  Clearly a normal
$S_{\bar{\pi}}$-subgroup of $\fix_{G_i}({\cal B})$ has relatively prime
order and index in $\fix_{G_k}({\cal B})$, $i = 1,2$.  Hence by
\cite[Theorem 1.1.13]{Hall1976}, an $S_{\bar{\pi}}$-subgroup of
$\fix_{G_i}({\cal B})$ is characteristic in $\fix_{G_i}({\cal
B})$, $i = 1,2$.  Whence an $S_{\bar{\pi}}$-subgroup of $\fix_{G_i}({\cal
B})$ is normal in $G_i$, $i = 1,2$.  Thus $G_i$ admits complete
block systems formed by the orbits of the $S_{\bar{\pi}}$-subgroup of
$\fix_{G_i}({\cal B})$, $i = 1,2$.  As each ${\cal C}_B$ is formed
by the orbits of the $S_{\bar{\pi}}$-subgroup of $\fix_{G_i}({\cal B})$
restricted to the block $B\in{\cal B}$, the orbits of the
$S_{\bar{\pi}}$-subgroup of $\fix_{G_i}({\cal B})$ form the complete block
system ${\cal C}$, $i = 1,2$.  Hence ${\cal C}$ is a block system
of $G_i$, $i = 1,2$, and so by Lemma \ref{blocktool}, ${\cal C}$
is a complete block system of $\la G_1, G_2\ra$.
\end{proof}

We now prove the main result of this section.

\begin{theorem}\label{main}
Let $n_1,\ldots,n_r$ be positive integers such that
$\gcd(n_i,n_j\cdot\varphi(n_j)) = 1$ if $i\not = j$, and $\pi_i$
be the set of distinct prime numbers dividing $n_i$.  Let $n =
n_1\cdots n_r$ and $m = \Omega(n)$.  If either
\begin{enumerate}
\item each $n_i$ is a prime-power, and $G_1,G_2$ are transitive nilpotent
groups of degree $n$ such that $\la G_1,G_2\ra$ is $m$-step
imprimitive, or
\item $G_1,G_2$ are transitive abelian groups of
degree $n$ such that $\la G_1,G_2\ra$ is $m$-step imprimitive,
\end{enumerate}
\noindent then there exists $\delta\in\la G_1,G_2\ra$ such that
$\la G_1,\delta^{-1}G_2\delta\ra = \Pi_{i=1}^rH_i$, where each
$H_i$ is a transitive solvable $\pi_i$-group of degree $n_i$.
\end{theorem}

\begin{proof}
Throughout the proof, if case (1) holds, we let $p_i$ be prime such that $n_i = p_i^{a_i}$, $a_i\ge 1$.  First suppose that case (1) holds and $r = 1$. Then $G_1\le\Pi_1$, $G_2\le \Pi_2$, where $\Pi_1,\Pi_2$
are Sylow $p_1$-subgroups of $\la G_1,G_2\ra$.  By a Sylow
Theorem, there exists $\delta\in\la G_1,G_2\ra$ such that
$\delta^{-1}G_2\delta\le\Pi_1$.  Then $\la
G_1,\delta^{-1}G_2\delta\ra$ is a $p_1$-group and so nilpotent.

In both cases, we proceed by induction on $m$.  Suppose that $m = 1$.  Then the only case that occurs is case (1) and $r = 1$.  The result then follows by arguments above.  Assume that the result is true for all $G_1$ and $G_2$ that satisfy the hypothesis with $\Omega(n) = m-1\ge 1$, and let $G_1,G_2\le S_n$ satisfy the hypothesis where
$\Omega(n) = m$. In case (1), by arguments above, we may assume that $r
\ge 2$. In case (2), if $r = 1$ then the result follows from Lemma \ref{piabelian}, so in any case we may assume without loss of generality that $r\ge 2$.  Let ${\cal B}_1$ be a complete block system of $\la
G_1,G_2\ra$ consisting of $p_i$ blocks of size $n/p_i$, where
$p_i\vert m_i$. Then ${\cal B}_1$ is a complete block system of
both $G_1$ and $G_2$. As both $G_1$ and $G_2$ are nilpotent,
$G_1/{\cal B}_1$ and $G_2/{\cal B}_1$ are nilpotent. We conclude
that both $G_1/{\cal B}_1$ and $G_2/{\cal B}_1$ are $p_i$-groups.
Hence there exists $\delta_1\in\la G_1,G_2\ra$ such that $\la
G_1,\delta^{-1}G_1\delta\ra/{\cal B}_1$ has order $p_i$. We thus
assume without loss of generality that $\la G_1,G_2\ra/{\cal B}_1$
has order $p_i$.

Let $\pi = \cup_{j\in J}\pi_j$ where $J = [r] - \{i\}$.  As $\vert
G_1/{\cal B}_1\vert = \vert G_2/{\cal B}_1\vert = p_i$, the
$S_\pi$-subgroups of $G_1$ and $G_2$ are contained in $\fix_{\la
G_1,G_2\ra}({\cal B}_1)$.  By Lemma \ref{bottomtool} and the
induction hypothesis, there exists $\delta_2\in\la G_1,G_2\ra$
such that $\la G_1,\delta_2^{-1}G_2\delta_2\ra$ admits a complete
block system ${\cal C}$ of $n_i$ blocks of size $n/n_i$ formed by
the $S_\pi$-subgroups of $G_1$ and $G_2$. We thus assume without
loss of generality that $\la G_1,G_2\ra$ admits ${\cal C}$ as a
complete block system.  Similarly, by Lemma \ref{bottomtool} and
the induction hypothesis, there exists $\delta_3\in\la G_1,G_2\ra$
such that $\la G_1,\delta_3^{-1}G_2\delta_3\ra$ admits a complete
block system ${\cal D}$ formed by the orbits of an
$S_{\pi_i}$-subgroup of $\fix_{G_k}({\cal B}_1)$, $k = 1,2$, (we
remark that if $n_i$ is prime, then ${\cal D}$ is trivial).  We
thus also assume without loss of generality that $\la G_1,G_2\ra$
admits ${\cal D}$ as well.

If $n_i\not = p_i$, then $\la G_1,G_2\ra/{\cal D}$ is $(m - (a_i -
1))$-step imprimitive, where $\Omega(n_i) = a_i$, and
$G_k/{\cal D}$ is nilpotent, $k = 1,2$. Hence by the induction
hypothesis there exists $\delta_4\in\la G_1,G_2\ra$ such that $\la
G_1,\delta_4^{-1}G_2\delta_4\ra/{\cal D}\le
P_i\times\Pi_{j=1,j\not = i}^rH_j'$, where $H_j'\le S_{n_j}$ is a
transitive solvable $\pi_j$-group, and $P_i$ is a $p_i$-group of
degree $p_i$. Then $\la G_1,\delta_4^{-1}G_2\delta_4\ra/{\cal D}$
admits a complete block system ${\cal E}'$ of $n/(n_i/p_i)$ blocks
of size $p_i$, so that $\la G_1,\delta_4^{-1}G_2\delta_4\ra$
admits a complete block system ${\cal E}$ of $n/n_i$ blocks of
size $n_i$.

If $n_i = p_i$, then as $\la G_1,G_2\ra$ is $m$-step imprimitive,
$\la G_1,G_2\ra$ admits a complete block system ${\cal B}_2$ such
that ${\cal B}_2\preceq{\cal B}_1$ and ${\cal B}_2$ consists of
$p_ip_j$ blocks of size $n/(p_ip_j)$ for some $p_j\vert n_j$ with
$j\not = i$. Then $G_1/{\cal B}_2$ and $G_2/{\cal B}_2$ are
nilpotent and transitive.  We conclude that $G_1/{\cal B}_2$ and
$G_2/{\cal B}_2$ are cyclic. By Theorem \ref{Paltool}, there
exists $\delta_3\in\la G_1,G_2\ra$ such that $\la
G_1,\delta_3^{-1}G_2\delta_3\ra/{\cal B}_2$ is cyclic.  We thus
assume without loss of generality that $\la G_1,G_2\ra/{\cal B}_2$
is cyclic.  Thus $\la G_1,G_2\ra/{\cal B}_2$ admits a complete
block system of $p_j$ blocks of size $p_i$, so that $\la
G_1,G_2\ra$ admits a complete block system ${\cal B}_1'$ of $p_j$
blocks of size $n/p_j$, and by Lemma \ref{mstep} $\fix_{\la
G_1,G_2\ra}({\cal B}_1')\vert_{B'}$ is $(m-1)$-step imprimitive
for every $B'\in{\cal B}_1'$.  Hence by Lemma \ref{bottomtool},
there exists $\delta_4\in\la G_1,G_2\ra$ such that $\la
G_1,\delta_4^{-1}G_2\delta_4\ra$ admits a complete block system
${\cal E}$ of $n/n_i$ blocks of size $n_i$ formed by the orbits of
an $S_{\pi_i}$-subgroup of $\fix_{G_k}({\cal B}_1')$, $k = 1,2$.
Hence regardless of the value of $n_i$, we may assume without loss
of generality that $\la G_1,G_2\ra$ admits ${\cal C}$ and ${\cal
E}$ as complete block systems.  As $\la G_1,G_2\ra$ admits a
complete block system ${\cal C}$ of $n_i$ blocks of size $n/n_i$,
$\la G_1,G_2\ra\le S_{n_i}\wr S_{n/n_i}$. As $\la G_1,G_2\ra$ also
admits ${\cal E}$ as a complete block system and $\gcd(n_i,n/n_i)
= 1$, we have that $\la G_1,G_2\ra\le S_{n/n_i}\wr S_{n_i}$. We
conclude that $\la G_1,G_2\ra\le (S_{n_i}\wr
S_{n/n_i})\cap(S_{n/n_i}\wr S_{n_i}) = S_{n_i}\times S_{n/n_i}$.
We now consider (1) and (2) separately.

(1)  By the induction hypothesis, we may, after a suitable
conjugation, assume that $\la G_1,G_2\ra/{\cal D}\le
\Pi_{j\in[r]-\{i\}}S_{m_j}$, so that $\la G_1,G_2\ra\le
\Pi_{j=1}^r S_{m_j}$.  The result then follows by inductively
applying a Sylow Theorem and then Lemma \ref{solproduct}.

(2)  By Lemma \ref{regularabelian} every complete block system is a normal
complete block system.  As ${\cal L}_{\la G_1,G_2\ra}$ is a
modular lattice by Lemma \ref{lattice}, it follows by the
Jordan-Dedekind Chain Condition \cite[pg. 119]{Hall1976} that all
finite chains between two elements have the same length.  As $\la
G_1,G_2\ra$ admits ${\cal E}$ as a complete block system, any
maximal chain between the complete block systems consisting of
singletons and the complete block system consisting of one block
that contain ${\cal E}$ must have length $m$ as $\la G_1,G_2\ra$
is normally $m$-step imprimitive. We conclude that $\la
G_1,G_2\ra/{\cal E}$ is $(m-a_i)$-step imprimitive, so by the
induction hypothesis we may assume after a suitable conjugation
that $\la G_1,G_2\ra/{\cal E}\le\Pi_{j=1,j\not = i}^rH_j$, where
$H_j\le S_{n_j}$ is a transitive solvable $\pi_j$-group.
Similarly, we may assume that $\la G_1,G_2\ra/{\cal C}\le H_i$,
where $H_i$ is a transitive solvable $\pi_i$-group.  As $C\cap E$ is a singleton, for every $C\in{\cal C}$, $E\in{\cal
E}$, we have that $\la G_1,G_2\ra\le\Pi_{j=1}^rH_j$. By Lemma
\ref{solproduct}, we may assume that $\la G_1,G_2\ra =
\Pi_{j=1}^rH_j$ as required.  The result then follows by
induction.
\end{proof}

It may be worthwhile restating Theorem \ref{main} (1) in the
following form:

\begin{corollary}\label{nilmain}
Let $n = p_1^{a_1}\cdots p_r^{a_r}$, the prime-power decomposition
of $n$, be such that $p_i\not\vert(p_j-1)$, $1\le i,j\le r$.  Let
$\Omega(n) = m$.  If $G_1,G_2$ are transitive nilpotent
groups of degree $n$ such that $\la G_1,G_2\ra$ is $m$-step
imprimitive, then there exists $\delta\in \la G_1,G_2\ra$ such
that $\la G_1,\delta^{-1}G_2\delta\ra$ is nilpotent.
\end{corollary}

\section{Solving Sets}

In this section, we further develop the terminology regarding solving sets as well as the characterizations of when a particular set is a solving set that will be needed for our main results.

\begin{definition}
Let $G$ be a group and define $g_L:G\rightarrow G$ by $g_L(x) =
gx$.  Let $G_L = \{g_L:g\in G\}$.  Then $G_L$ is the {\bf
left-regular representation of $G$}.  We define a {\bf Cayley
object of $G$} to be a combinatorial object $X$ (e.g. digraph,
graph, design, code) such that $G_L\le\Aut(X)$, where $\Aut(X)$ is
the {\bf automorphism group of $X$} (note that this implies that
the vertex set of $X$ is in fact $G$).  If $X$ is a Cayley object
of $G$ in some class ${\cal K}$ of combinatorial objects with the
property that whenever $Y$ is another Cayley object of $G$ in
${\cal K}$, then $X$ and $Y$ are isomorphic if and only if they
are isomorphic by a group automorphism of $G$, then we say that
$X$ is a {\bf CI-object of $G$ in ${\cal K}$}.  If every Cayley
object of $G$ in ${\cal K}$ is a CI-object of $G$ in ${\cal K}$,
then we say that $G$ is a {\bf CI-group with respect to ${\cal
K}$}.  If $G$ is a CI-group with respect to every class of
combinatorial objects, then $G$ is a {\bf CI-group}.
\end{definition}

\begin{definition}
Let $G$ be a finite group.  We say that $S\subseteq S_G$ is a {\bf
solving set for a Cayley object $X$} in a class of Cayley objects
${\cal K}$ if for every $X'\in {\cal K}$ such that $X\cong X'$,
there exists $s\in S$ such that $s(X) = X'$, $s(1_G) = 1_G$ for every $s\in S$, and $\Aut(G)\le S$. We
say that $S\subseteq S_G$ is a {\bf solving set for a class ${\cal
K}$ of Cayley objects of $G$} if whenever $X,X'\in {\cal K}$ are
Cayley objects of $G$ and $X\cong X'$, then $s(X) = X'$ for some
$s\in S$, and $s(1_G) = 1_G$ for every $s\in S$, and $\Aut(G)\le S$.  Finally, a set
$S$ is a {\bf solving set for $G$} if whenever $X,X'$ are isomorphic
Cayley objects of $G$ in any class ${\cal K}$ of combinatorial objects,
then $s(X) = X'$ for some $s\in S$, $s(1_G) = 1_G$ for all $s\in S$, and $\Aut(G)\le S$.
\end{definition}

\begin{remark}
Note that the definition of a solving set given above differs from
those in \cite{Muzychuk1999} and \cite{Dobson2003a}, as here, to simplify both the statements of results and their proofs, we insist that every
element of the solving set fixes $1_G$.  It is easy to see that $\alpha^{-1}G_L\alpha = G_L$ for every $\alpha\in\Aut(G)$, so the image of a Cayley object of $G$ under a group automorphism of $G$ is a Cayley object of $G$.  That is, in order to test for isomorphism, automorphisms of $G$ must be considered.  However, it is not always the case that {\it every} automorphism of $G$ needs to be considered when testing for isomorphism.  For example, Cayley graphs of cyclic groups of order $n$ each have an automorphism $x\to -x$ that is also a group automorphism of $\Z_n$, and so the image of a Cayley graph under this automorphism of $\Z_n$ is itself.  So, while our definition of a solving set is convenient for this paper, it does not always capture the idea behind a solving set (i.e. that it should be as small as possible) exactly, but will only necessarily include extra automorphism of $G$ (which could then be excluded).  Also note that in
\cite{Muzychuk1999} and \cite{Dobson2003a}, solving sets were only defined for
abelian groups.
\end{remark}

Let $X$ be a Cayley object of $G$ in ${\cal K}$.  We define a
{\bf CI-extension of $G$ with respect to $X$}, denoted by
$\CI(G,X)$, to be a set of permutations in $S_G$ that each
fix $1_G$ and whenever $\delta\in S_G$ such that
$\delta^{-1}G_L\delta\le\Aut(X)$, then there exists $v\in\Aut(X)$
such that $v^{-1}\delta^{-1}G_L\delta v = t^{-1}G_Lt$ for some
$t\in \CI(G,X)$.

\begin{lemma}\label{existence}
Let $G$ be a finite group, and $X$ a Cayley object of $G$ in some
class ${\cal K}$ of combinatorial objects.  Then $\CI(G,X)$ exists.
\end{lemma}

\begin{proof}
To show existence, we only need show that there is a set of
permutations $T$ in $S_G$ such that whenever $\delta\in S_G$ such
that $\delta^{-1}G_L\delta\le\Aut(X)$ and $v\in\Aut(X)$, then
$v^{-1}\delta^{-1}G_L\delta v = t^{-1}G_Lt$ for some $t\in
T$ and $t(1_G) = 1_G$.  This follows almost immediately.  As
$\delta v$ is a permutation, there exists $g\in G$ such that
$\delta v(1_G) = g$.  Let $t_{\delta v} = g^{-1}_L\delta v$.  Then
$t_{\delta v}(1_G) = g^{-1}_L\delta v(1_G) = g^{-1}_L(g) = g^{-1}
g = 1_G$. Furthermore, $t_{\delta v}^{-1}G_Lt_{\delta v} =
v^{-1}\delta^{-1}g_LG_Lg_L^{-1}\delta v =
v^{-1}\delta^{-1}G_L\delta v$, and existence is established with $T = \{t_{\delta v}:\delta^{-1}G_L\delta\le\Aut(X), v\in \Aut(X)\}$.
\end{proof}

Note for $X$ a Cayley object of $G$ in ${\cal K}$,
$\CI(G,X)$ is not unique as if $T$ is CI-extension of $X$ with
respect to $G$, then for $\alpha\in\Aut(G)$, $\{\alpha t:t\in T\}$
is also a CI-extension of $X$ with respect to $G$.  The following result shows the importance of
$\CI(G,X)$, as if $\CI(G,X)$ is known, then the isomorphism problem
is solved.

\begin{lemma}\label{contool1}
Let $G$ be a finite group, and $X$ a Cayley object of $G$ in some
class ${\cal K}$ of combinatorial objects.  Then the following are
equivalent:
\begin{enumerate}
\item $S = \{\alpha t:\alpha\in\Aut(G), t\in T \}$ is a solving set for $X$,
\item $T$ is a $\CI(G,X)$.
\end{enumerate}
\end{lemma}

\begin{proof}
1) implies 2).   Let $\delta\in S_G$ such that
$\delta^{-1}G_L\delta\le\Aut(X)$.  Then $\delta(X)$ is a Cayley
object of $G$ in ${\cal K}$  as $\Aut(\delta(X)) = \delta\Aut(X)\delta^{-1}\ge G_L$.  As $S$ is a solving set for $X$,
$\delta(X) = s(X)$ for some $s\in S$, and $s = \alpha t$ for some
$\alpha\in\Aut(G)$ and $t\in T$.  Thus $v =
\delta^{-1}s\in\Aut(X)$. Then
\begin{eqnarray*}
v^{-1}\delta^{-1}G_L\delta v & = & s^{-1}\delta \delta^{-1}G_L\delta\delta^{-1}s\\
  & = & s^{-1}G_Ls = t^{-1}\alpha^{-1}G_L\alpha t\\
  & = & t^{-1}G_Lt
\end{eqnarray*}

\noindent and $T$ is a $\CI(X,G)$.

2) implies 1).  Let $X$ and $X'$ be isomorphic Cayley objects of
$G$ in ${\cal K}$.  Then there exists $\delta\in S_G$ such that
$\delta(X) = X'$.  As $G_L\le\Aut(X')$,
$\delta^{-1}G_L\delta\le\Aut(X)$. As $T$ is a $\CI(X,G)$, there exists
$t\in T$ and $v\in\Aut(X)$ such that
$v^{-1}\delta^{-1}G_L\delta v = t^{-1}G_Lt$.  Hence
$tv^{-1}\delta^{-1}G_L\delta v t^{-1} = G_L$. As $G_L$ is
transitive, there exists $h\in G$ such that $h_L\delta
vt^{-1}(1_G) = 1_G$.  Clearly then $tv^{-1}\delta^{-1}h_L^{-1}G_L
h_L\delta v t^{-1} = G_L$ so that $h_L\delta vt^{-1}$ normalizes $G_L$ and fixes $1_G$.  By \cite[Corollary 4.2B]{DixonM1996}, we have that $h_L\delta vt^{-1} = \alpha\in\Aut(G)$. Then $\alpha t
v^{-1}\delta^{-1}h_L^{-1} = (1_G)_L$ and
%$$(h_L\delta v t^{-1})^{-1}g_Lh_L\delta v t^{-1} = \alpha^{-1}g_L\alpha$$
%\noindent for every $g\in G$.  Thus $\alpha t
%v^{-1}\delta^{-1}h_L^{-1}$ is contained in the center of $G_L$. As
%the centralizer of $G_L$ in $S_G$ is $G_R$ (that
%$\C_{S_G}(G_L)\cong G$ follows from \cite[Theorem 4.2A (iii)]{DixonM1996}.
%It is then a simple computation to show that $\C_{S_G}(G_L) =
%G_R$), we have that $\alpha tv^{-1}\delta^{-1}h_L^{-1}\in G_R$. As
%$\alpha(1_G) = 1_G$ and $h_L\delta vt^{-1}(1_G) = 1_G$, we have
%that $\alpha tv^{-1}\delta^{-1}h_L^{-1}(1_G) = 1_G$. As the only
%element of $G_R$ with a fixed point is $(1_G)_R = (1_G)_L$, we
%have that $\alpha tv^{-1}\delta^{-1}h_L^{-1} = (1_G)_L$.  Thus
$\alpha t = h_L\delta v$.  Then $\alpha t(X) = h_L\delta v(X) =
h_L\delta(X) = h_L(X') = X'$.
\end{proof}

The following result shows that if a solving set for $X$ has been found, then some $\CI(G,X)$ has also been found.

\begin{lemma}\label{CIX}
Let $G$ be a group, $X$ a Cayley object of $G$, and $S$ a solving set for $X$.  Define an equivalence relation $\equiv$ on $S$ by $s_1\equiv s_2$ if and only if $s_1 = \alpha s_2$ for some $\alpha\in\Aut(G)$.  Let $T$ be a set consisting of one representative from each equivalence class of $\equiv$.  Then $T$ is a $\CI(G,X)$.
\end{lemma}

\begin{proof}
It is straightforward to show that $\equiv$ is indeed an equivalence relation.  Choose a $T$ as is given in the statement.   Let $X'$ be a Cayley object of $G$ isomorphic to $X$ with $\delta:X\to X'$ an isomorphism.  Then $\delta^{-1}G_L\delta\le\Aut(X)$.  Also, as $S$ is a solving set for $X$, there exists $s\in S$ such that $s(X) = X'$ so that $v = \delta^{-1}s\in\Aut(X)$.  Let $t\in T$ such that $t\equiv s$ so that $\alpha t = s$ for some $\alpha\in\Aut(G)$.  Then $v^{-1}\delta^{-1}G_L\delta v = t^{-1}\alpha^{-1}G_L\alpha t = t^{-1}G_Lt$.
\end{proof}

Let ${\cal K}$ be a class of combinatorial objects, and $G$ a
group.  We define a {\bf CI-extension of $G$ with respect to
${\cal K}$}, denoted by $\CI(G,{\cal K})$, to be a set of
permutations in $S_G$ that each fix $1_G$ and whenever $X\in{\cal
K}$ is a Cayley object of $G$ and $\delta\in S_G$ such that
$\delta^{-1}G_L\delta\le\Aut(X)$, then there exists $t\in
\CI(G,{\cal K})$ and $v\in\Aut(X)$ such that
$v^{-1}\delta^{-1}G_L\delta v = t^{-1}G_Lt$. The proofs of the
following results are straightforward.

\begin{lemma}\label{contool2}
Let $G$ be a finite group, and ${\cal K}$ a class of combinatorial
objects.  Then the following are equivalent:
\begin{enumerate}
\item $S = \{\alpha t:\alpha\in\Aut(G), t\in T\}$ is a solving set for $G$ in ${\cal K}$,
\item $T$ is a $\CI(G,{\cal K})$.
\end{enumerate}
\end{lemma}

\begin{lemma}\label{CIK}
Let $G$ be a group, ${\cal K}$ a class of combinatorial objects, and $S$ a solving set for $G$ in ${\cal K}$.  Define an equivalence relation $\equiv$ on $S$ by $s_1\equiv s_2$ if and only if $s_1 = \alpha s_2$ for some $\alpha\in\Aut(G)$.  Let $T$ be a set consisting of one representative from each equivalence class of $\equiv$.  Then $T$ is a $\CI(G,{\cal K})$.
\end{lemma}

Let $G$ be a finite group.  We define a {\bf CI-extension of
$G$}, denoted by $\CI(G)$, to be a set of permutations in
$S_G$ that each fix $1_G$ and whenever $X\in{\cal K}$ is a Cayley
object of $G$ in some class ${\cal K}$ of combinatorial objects,
and $\delta\in S_G$ such that $\delta^{-1}G_L\delta\le\Aut(X)$,
then there exists $t\in \CI(G)$ and $v\in\la G_L,\delta^{-1}G_L\delta\ra$ such that
$v^{-1}\delta^{-1}G_L\delta v = t^{-1}G_Lt$. Repeated application of Lemma \ref{existence} for every combinatorial object $X$ in every class ${\cal K}$ of combinatorial objects shows that $\CI(G)$ exists.  The proofs of the
following results are straightforward.

\begin{lemma}\label{contool3}
Let $G$ be a finite group.  Then the following are equivalent:
\begin{enumerate}
\item $S = \{\alpha t:\alpha\in\Aut(G), t\in T\}$ is a solving set for $G$,
\item $T$ is a $\CI(G)$.
\end{enumerate}
\end{lemma}

\begin{lemma}\label{CIG}
Let $G$ be a group, and $S$ a solving set for $X$.  Define an equivalence relation $\equiv$ on $S$ by $s_1\equiv s_2$ if and only if $s_1 = \alpha s_2$ for some $\alpha\in\Aut(G)$.  Let $T$ be a set consisting of one representative from each equivalence class of $\equiv$.  Then $T$ is a $\CI(G)$.
\end{lemma}

\section{Applications}

At the present time, the isomorphism problem has not been solved for any nilpotent group that is not abelian in any class of combinatorial objects, so there are not at this time any applications for Theorem \ref{main} (1) (although as soon as the isomorphism problem has been solved for any nonabelian $p$-group in certain classes of combinatorial objects, such as color digraphs, that will change immediately).  We do though have an application of Theorem \ref{main} (2) which will not only provide new examples of CI-groups with respect to color digraphs, but also illustrate how Theorem \ref{main} (2) generalizes the main result of \cite{Dobson2003a}.

%For clarity, we now give the basic definitions concerning the CI-problem, some of which have already been stated.

%\begin{defin}
%Let $G$ be a group and define $g_L:G\rightarrow G$ by $g_L(x) =
%gx$.  Let $G_L = \{g_L:g\in G\}$.  Then $G_L$ is the {\bf
%left-regular representation of $G$}.  We define a {\bf Cayley
%object of $G$} to be a combinatorial object $X$ (e.g. digraph,
%graph, design, code) such that $G_L\le\Aut(X)$, where $\Aut(X)$ is
%the {\bf automorphism group of $X$} (note that this implies that
%the vertex set of $X$ is in fact $G$).  If $X$ is a Cayley object
%of $G$ in some class ${\cal K}$ of combinatorial objects with the
%property that whenever $Y$ is another Cayley object of $G$ in
%${\cal K}$, then $X$ and $Y$ are isomorphic if and only if they
%are isomorphic by a group automorphism of $G$, then we say that
%$X$ is a {\bf CI-object of $G$ in ${\cal K}$}.  If every Cayley
%object of $G$ in ${\cal K}$ is a CI-object of $G$ in ${\cal K}$,
%then we say that $G$ is a {\bf CI-group with respect to ${\cal
%K}$}.  If $G$ is a CI-group with respect to every class of
%combinatorial objects, then $G$ is a {\bf CI-group}.
%\end{defin}

The following result weakens the hypothesis (replacing normally
$s$-step imprimitive with $s$-step imprimitive) of \cite[Theorem
16]{Dobson2003a} and generalizes this result from abelian to nilpotent
groups.

\begin{theorem}\label{bigtool}
Let $n_1,\ldots,n_r$ be positive integers such that
$\gcd(n_i,n_j\cdot\varphi(n_j)) = 1$ if $i\not = j$, and $\pi_i$
be the set of distinct prime numbers dividing $n_i$.  Let $n =
n_1\cdots n_r$, $\Omega(n) = m$, and $G$ a nilpotent group of degree $n$.  Let $G = \Pi_{i=1}^rN_i$ where each $N_i$ is a $\pi_i$-subgroup of $G$, and $S(i)$ a solving set for $N_i$.  If
\begin{enumerate}
\item each $n_i$ is prime-power or $G$ is abelian, and
\item whenever $\delta\in S_G$ there exists $\phi\in\la G_L,\delta^{-1}G_L\delta\ra$ such that $\la G_L,\phi^{-1}\delta^{-1}G_L\phi\delta\ra$ is $m$-step imprimitive,
\end{enumerate}
\noindent then $\Pi_{i=1}^rS(i)$ is a solving set for $G$.
\end{theorem}

\begin{proof}
Let $\delta\in S_G$.  By the hypothesis, we have that there exists $\phi\in\la G_L,\delta^{-1}G_L\delta\ra$ such that $\la G_L,\phi^{-1}\delta^{-1}G_L\delta\phi\ra$ is $m$-step imprimitive.  By Theorem \ref{main}, there exists
$\omega\in\la G_L,\phi^{-1}\delta^{-1}G_L\delta\phi\ra$ such
that $L = \la G_L, \omega^{-1}\phi^{-1}\delta^{-1}G_L\delta\phi\omega\ra =
\Pi_{i=1}^rL_i$, where each $L_i$ is a transitive $\pi_i$-group of
degree $n_i$, $1\le i\le r$.  Let $\CI(N_i)$ be a CI-extension of $N_i$ as given by Lemma \ref{CIG}.  As $S(i)$ is a solving set for $N_i$, by Lemma \ref{contool3} there exists $t_i\in \CI(N_i)$ and
$v_i\in L_i$ such that $v_i^{-1}((\omega^{-1}\phi^{-1}\delta^{-1}G_L\delta\phi\omega)/{\cal
C}_i)v_i = t_i^{-1}(G_L/{\cal C}_i)t_i$, where ${\cal C}_i$ is the complete block system of $L$ formed by the orbits of $\Pi_{j=1,j\not = i}^rL_i$.  Let $t = (t_1,\ldots,t_r)$ and $v = (v_1,\ldots,v_r)$. Note that $t$ fixes $1_G$.  As $L = \Pi_{i=1}^rL_i$, we have that $v\in L$ and $t^{-1}G_Lt =
v^{-1}\omega^{-1}\phi^{-1}\delta^{-1}G_L\delta\phi\omega v$. By definition, $\Pi_{i=1}^r\CI(N_i)$ is a CI-extension of $G$, and so by Lemma \ref{contool3}, $S = \{\alpha t:t\in\Pi_{i=1}^r\CI(N_i), \alpha\in\Aut(G)\}$ is a solving set for $G$.  Finally, it is not difficult to see that $\Aut(G) = \Pi_{i=1}^r\Aut(N_i)$ as $G$ is nilpotent and if $i\not = j$ then $\gcd(n_i,n_j) = 1$, and so $\alpha = (\alpha_1,\alpha_2,\ldots,\alpha_r)$, $\alpha_i\in\Aut(N_i)$.  Then $\alpha t = (\alpha_1 t_1,\alpha_2 t_2,\ldots,\alpha_r t_r)$, $\alpha_i\in\Aut(N_i)$ and $t_i\in \CI(N_i)$.  Thus $\alpha t\in\Pi_{i=1}^rS(i)$, $S\le\Pi_{i=1}^r S(i)$, and $\Pi_{i=1}^rS(i)$ is a solving set for $G$.
%
%Now let $u\in v^{-1}\omega^{-1}\phi^{-1}\delta^{-1}G_L\delta\phi\omega v$ such that $\delta\phi\omega vu(1_G) = %1_G$, and $\gamma = \phi\omega vu$.  Then $\gamma\in\la G_L,\delta^{-1}G_L\delta\ra$, and $\delta\gamma$ fixes %$1_G$.
%Also, $t\gamma^{-1}\delta^{-1}G_L\delta\gamma t^{-1} = G_L$, and so by \cite[Corollary 4.2B]{DixonM1996} it follows %that $\delta\gamma t^{-1}\in\Aut(G)\cdot G_L$.  As both $t$ and $\delta\gamma$ fix $1_G$, we have that $\delta\gamma %t^{-1} = \alpha\in\Aut(G)$, and so $\delta\gamma = \alpha t$.  As $\delta\gamma(X) = X'$, $\alpha t(X) = X'$.  %Finally, it is not difficult to see that $\Aut(G) = \Pi_{i=1}^r\Aut(P_i)$ as $G$ is nilpotent, and so $\alpha = %(\alpha_1,\alpha_2,\ldots,\alpha_r)$, $\alpha_i\in\Aut(P_i)$.  Then $\alpha t = (\alpha_1 t_1,\alpha_2 %t_2,\ldots,\alpha_r t_r)$, $\alpha_i\in\Aut(P_i)$ and $t_i\in \CI(P_i)$.  Thus $\alpha t\in\Pi_{i=1}^rS(i)$ and the %result follows.
\end{proof}

\begin{definition}
{\rm Let $\Omega$ be a set.  A {\bf $k$-ary relational structure} on
$\Omega$ is an ordered pair $(\Omega, U)$, where $U\subseteq \Omega^k = \Pi_{i=1}^k\Omega$.  A group
$G\le S_{\Omega}$ is called {\bf $k$-closed} if $G$ is the
intersection of the automorphism groups of some set of $k$-ary
relational structures.  The {\bf $k$-closure of $G$}, denoted
$G^{(k)}$, is the intersection of all $k$-closed subgroups of
$S_{\Omega}$ that contain $G$.}
\end{definition}

Note that a $2$-closed group is the automorphism group of a color digraph.  The following result of Kalu\v znin and Klin \cite{KaluzninK1976} will prove useful.

\begin{lemma}\label{product}
Let $G\le S_X$ and $H\le S_Y$.  Let $G\times H$ act canonically on
$X\times Y$.  Then $(G\times H)^{(k)} = G^{(k)}\times H^{(k)}$ for
every $k\ge 2$.
\end{lemma}

If, in Theorem \ref{bigtool}, ${\cal K}$ is the class of $k$-ary
relational structures, and the groups $L/{\cal C}_i$ are as in the
proof of Theorem \ref{bigtool}, then by Lemma \ref{product} we may assume that each
$L/{\cal C}_i$ is $k$-closed (although there is no reason to believe that each $L/{\cal C}_i$ is a $\pi_i$-subgroup - but this fact is not used in the proof of Theorem \ref{bigtool}).  Proceeding as in Theorem
\ref{bigtool} and applying Lemma \ref{contool2} instead of Lemma
\ref{contool3}, we have the following result.

\begin{corollary}\label{kcor}
Let $n_1,\ldots,n_r$ be positive integers such that
$\gcd(n_i,n_j\cdot\varphi(n_j)) = 1$ if $i\not = j$, and $\pi_i$
be the set of distinct prime numbers dividing $n_i$.  Let $n =
n_1\cdots n_r$, $\Omega(n) = m$, and $G$ a nilpotent group of degree $n$.  Let $G = \Pi_{i=1}^rN_i$ where each $N_i$ is a $\pi_i$-subgroup of $G$, and $S(i)$ a solving set for $N_i$ in the class of $k$-ary relational structures.  If
\begin{enumerate}
\item each $n_i$ is prime or $G$ is abelian, and
\item whenever $\delta\in S_G$ there exists $\phi\in\la G_L,\delta^{-1}G_L\delta\ra$ such that $\la G_L,\phi^{-1}\delta^{-1}G_L\phi\delta\ra$ is $m$-step imprimitive,
\end{enumerate}
\noindent then $\Pi_{i=1}^rS(i)$ is a solving set for $G$ in the class of $k$-ary relational structures.
\end{corollary}

%\begin{cor}
%Let $m$ be a positive integer with $\gcd(m,\varphi(m)) = 1$ and $m
%= p_1p_2\cdots p_r$ be the prime power decomposition of $m$. Let
%$n = p_1^{a_1}p_2^{a_2}\cdots p_r^{a_r}$, where $a_\ell\ge 1$ is a
%positive integer.  Let $X$ be $k$-ary Cayley object of a nilpotent
%group $G = \Pi_{i=1}^rP_i$ of order $n$, where $P_i$ is a Sylow
%$p_i$-subgroup of $G$.  Let $S(i)$ be the solving set for $P_i$ in
%the class of $k$-ary relational structures. If, whenever
%$\delta\in S_G$ such that $\delta^{-1}G_L\delta\le\Aut(X)$ there
%exists $\omega\in\Aut(X)$ such that $\la
%G_L,\omega^{-1}\delta^{-1}G_L\delta\omega\ra$ is $s$-step
%imprimitive, then there exists a solving set for $X$ that is
%contained in $\Pi_{i=1}^rS(i)$.
%\end{cor}

%Babai \cite{Babai1977} characterized the CI-property in the following
%manner.

%\begin{lem}\label{CIso}
%For a Cayley object $X$ of $G$ in ${\cal K}$ the following are
%equivalent:
%\begin{enumerate}
%\item $X$ is a CI-object,
%\item given a permutation $\phi \in S_G$ such that $\phi^{-1}G_L\phi\le \Aut(X)$,
%$G_L$ and $\phi^{-1}G_L\phi$ are conjugate in $\Aut(X)$.
%\end{enumerate}
%\end{lem}

%We remark that in \cite{Babai1977}, similar characterizations of CI-groups with respect to classes of combinatorial %objects and to all classes of combinatorial objects were also give.

We now give the promised application of Theorem \ref{main} (2) which gives new CI-groups with respect to Cayley color digraphs of a particular group.  Using the results in \cite{Dobson2003a}, this result could be obtained but only in the special cases where the following additional arithmetic conditions hold: $p$ does not divide $q - 1$, and each $n_i$ is prime.  Before proceeding, we need a preliminary lemma.

\begin{lemma}\label{bigblocks}
Let $n$ be a positive integer with a prime divisor $p\vert n$ such that $n/p < p$.  If $G$ is a regular group of order $n$, and $\phi\in S_n$, then there exists $\delta\in \la G,\phi^{-1}G\phi\ra$ such that $\la G,\delta^{-1}\phi^{-1}G\phi\delta\ra$ admits a normal complete block system with blocks of size $p$.
\end{lemma}

\begin{proof}
First observe that as $n/p < p$, $G$ must have a unique cyclic Sylow $p$-subgroup $P$ of order $p$.  So $P\tl G$ and $G$ admits a normal complete block system ${\cal B}$ consisting of blocks of size $p$, formed by the orbits of $P$.  Furthermore, a Sylow $p$-subgroup of $S_n$ is of the form $1_{S_{n/p}}\wr\Z_p$, so we see that $\la P\vert_B:B\in{\cal B}\ra$ is a Sylow $p$-subgroup of $S_n$.  By a Sylow theorem, there exists $\delta\in \la G,\phi^{-1}G\phi\ra$ such that $\delta^{-1}\phi^{-1}G\phi\delta\le \la P\vert_B:B\in{\cal B}\ra$.  Let $P_1$ be the largest subgroup of $\la G,\delta^{-1}\phi^{-1}G\phi\delta\ra$ contained in $\la P\vert_B:B\in{\cal B}\ra$.  Then $G$ normalizes $P_1$ as does $\delta^{-1}\phi^{-1}G\phi\delta$.  Hence the orbits of $P_1$, which is ${\cal B}$, is a complete block system of $\la G,\delta^{-1}\phi^{-1}G\phi\delta\ra$ and the result follows.
\end{proof}

\begin{theorem}\label{app}
Let $p$ and $q$ be distinct primes with $p^2 < q$, and $q_1,\ldots,q_r$ distinct primes such that $qp^2 < q_1$ and $qp^2q_1\cdots q_i < q_{i+1}$, $1\le i\le r - 1$.  Let $m = q_1\cdots q_r$, $n_0 = p^2q$, and $n_1,\ldots,n_s$ divisors of $m$ such that $n_1\cdots n_s = m$. If $\gcd(n_i,n_j\cdot\varphi(n_j)) = 1$ then $\Z_q\times\Z_p^2\times\Z_m$ is a CI-group with respect to digraphs.
\end{theorem}

\begin{proof}
Let $\Gamma$ be a Cayley color digraph of $G = \Z_q\times\Z_p^2\times\Z_m$, and $\phi\in S_G$ such that $\phi^{-1}G_L\phi\le\Aut(\Gamma)$.  We first show by induction on $r$ that $\la G_L,\phi^{-1}G_L\phi\ra$ is $s$-step imprimitive, where $s = r + 3$.  If $r = 0$, then as $p^2 < q$ by Lemma \ref{bigblocks} we may assume after an appropriate conjugation that $\la G,\phi^{-1}G\phi\ra$ admits a (normal) complete block system ${\cal B}_q$ with blocks of size $q$. Then $G_L/{\cal B}_q$ and $\phi^{-1}G_L\phi/{\cal B}_q$ are contained in conjugate Sylow $p$-subgroups of $\la G_L,\phi^{-1}G_L\phi\ra/{\cal B}_q$, and so after another appropriate conjugation we may assume that $\la G_L,\phi^{-1}G_L\phi\ra/{\cal B}_q$ is a $p$-group.  The center of $\la G_L,\phi^{-1}G_L\phi\ra/{\cal B}_q$ contains an element of order $p$ whose orbits give a complete block system ${\cal B}_p$ consisting of blocks of size $p$.  Then ${\cal B}_p$ induces a complete block system ${\cal B}_q\prec {\cal B}_{pq}$ of $\la G_L,\phi^{-1}G_L\phi\ra$ with blocks of size $pq$.  Hence $\la G_L,\phi^{-1}G_L\phi\ra$ is $3$-step imprimitive.  Assume that $\la G_L,\phi^{-1}G_L\phi\ra$ is $s$-step imprimitive when $r\ge 0$ and let $G_L$ be such that $m$ is a product of $r + 1$ primes.  Again, after an application of Lemma \ref{bigblocks}, we may assume that $\la G_L,\phi^{-1}G_L\phi\ra$ admits a complete block system ${\cal B}_{r + 1}$ with blocks of size $q_{r + 1}$.  By induction we may assume that $\la G_L,\phi^{-1}G_L\phi\ra/{\cal B}_{q_{r+1}}$ is $(r + 3)$-step imprimitive.  It is then not difficult to see that $\la G_L,\phi^{-1}G_L\phi\ra$ is $(r + 4)$-step imprimitive.  By induction, we then have $\la G_L,\phi^{-1}G_L\phi\ra$ is $(r + 3)$-step imprimitive as required.

Write $G_L = \Pi_{i=0}^sG_i$, where $G_i\le G_L$ is of order $n_i$.  As $G_i$ is a CI-group with respect to Cayley color digraphs by \cite{KovacsM2009} if $i = 0$ and by \cite{Muzychuk1997} otherwise, by Corollary \ref{kcor}, we see that if $S$ is a solving set of $G$ in the class of color digraphs then $S \subseteq \Pi_{i=1}^r\Aut(G_i)\le\Aut(G)$.  Hence $G$ is a CI-group with respect to color digraphs.
\end{proof}




\end{document}

